Author's response to reviews

Title: Consensus building to improve the physical health of people with severe mental illness: A qualitative outcome mapping study

Authors:

   Carolyn Ehrlich (c.ehrlich@griffith.edu.au)
   Elizabeth Kendall (e.kendall@griffith.edu.au)
   Nicolette Frey (n.frey@griffith.edu.au)
   Michelle Denton (m.denton@griffith.edu.au)
   Steve Kisely (s.kisely@uq.edu.au)

Version: 6  Date: 28 January 2015

Author's response to reviews: see over
28\textsuperscript{th} of January 2015

Dear Sir/Madam

“Consensus building to improve the physical health of people with severe mental illness: A qualitative outcome mapping study?”

The authors are grateful to the Journal for their feedback. In response, please find our paper “Consensus building to improve the physical health of people with severe mental illness: A qualitative outcome mapping study?” attached with the following changes incorporated as per instructed:

1. Authors contributions has been moved to the correct place in the manuscript

Please do not hesitate to contact me should there be any other changes required. We look forward to hearing from you.

Yours Sincerely,

Carolyn Ehrlich, PhD
Population and Social Health Research Program
Griffith University

c.ehrlich@griffith.edu.au; Ph: +61 7 3382 1296
Fax: +61 6 3382 1414